Author's response to reviews

Title: Interventricular septum hematoma during cineventriculography

Authors:

Andrea Grohmann (andrea.grohmann@charite.de)
Thomas Elgeti (thomas.elgeti@charite.de)
Stephan Eddicks (stephan.eddicks@charite.de)
Fabian Knebel (fabian.knebel@charite.de)
Wolfgang Rutsch (wolfgang.rutsch@charite.de)
Christoph Melzer (christoph.melzer@charite.de)
Gert Baumann (gert.baumann@charite.de)
Adrian C. Borges (adrian.borges@charite.de)

Version: 2 Date: 11 October 2007

Author's response to reviews:

Dear Eugenio Picano,

we would like to re-submit our case report describing a rare case with "Interventricular septum hematoma during cineventriculography". We have changed the paper according to the reviewer's suggestions. We hope that the paper in its current form meets the criteria for publication in "Cardiovascular Ultrasound".

Yours sincerely,

Andrea Grohmann
Fabian Knebel
Adrian C. Borges